Author's response to reviews

Title: Temporomandibular joint disc repositioning using anchor: an effective method to treat internal derangement in stage to

Authors:

ShanYong zhang (zhangshanyong@126.com)
XiuMing Liu (X.M.Liu@yahoo.com.cn)
XiuJuan Yang (X.J.Yang@126.com)
Chi Yang (yangchi63@hotmail.com)
MinJie Chen (M.J.Chen@126.com)
Majd S Haddad (M.S.haddad@uiowa.edu)
ZhuoZhi Chen (Z.Z.Chen@hotmail.com)

Version: 2 Date: 7 September 2009

Author's response to reviews: see over
Dear editor,

Thank you very much for publishing my paper in your Journal this year. Now I have submitted another paper entitled “Temporomandibular joint disc repositioning using anchor: an effective method to treat internal derangement in stage □ to □” to *BMC Musculoskelet Disord*. The manuscript has been read and approved by all authors. Also, the work is original and the paper is not under consideration for publication in any other journals. If you need further information about this paper, please let me know. I am looking forward to hearing from you.

Best wishes,
Yours Sincerely

ShanYong ZHANG, MD, DDS
Department of Oral and Maxillofacial Surgery
Ninth People’s Hospital
School of Medicine
Shanghai Jiaotong University
Zhi-Zao-Ju Rd., No. 639
Shanghai 200011, P.R.China
zhangshanyong@126.com